

Clonal expansion of early to mid-life mitochondrial DNA point mutations drives mitochondrial dysfunction during human ageing

Article

Published Version

Creative Commons: Attribution 3.0 (CC-BY)

Open Access

Greaves, L. C., Nooteboom, M., Elson, J. L., Tuppen, H. A. L., Taylor, G. A., Commane, D. M., Arasaradnam, R. P., Khrapko, K., Taylor, R. W., Kirkwood, T. B. L., Mathers, J. C. and Turnbull, D. M. (2014) Clonal expansion of early to mid-life mitochondrial DNA point mutations drives mitochondrial dysfunction during human ageing. PLoS Genetics, 10 (18). e1004620. ISSN 1553-7404 doi: 10.1371/journal.pgen.1004620 Available at https://centaur.reading.ac.uk/38065/

It is advisable to refer to the publisher's version if you intend to cite from the work. See <u>Guidance on citing</u>.

To link to this article DOI: http://dx.doi.org/10.1371/journal.pgen.1004620

Publisher: Public Library of Science

All outputs in CentAUR are protected by Intellectual Property Rights law, including copyright law. Copyright and IPR is retained by the creators or other



copyright holders. Terms and conditions for use of this material are defined in the End User Agreement.

www.reading.ac.uk/centaur

CentAUR

Central Archive at the University of Reading

Reading's research outputs online

Clonal Expansion of Early to Mid-Life Mitochondrial DNA Point Mutations Drives Mitochondrial Dysfunction during Human Ageing



Laura C. Greaves^{1,2*}, Marco Nooteboom², Joanna L. Elson^{3,4}, Helen A. L. Tuppen², Geoffrey A. Taylor², Daniel M. Commane⁵, Ramesh P. Arasaradnam⁵, Konstantin Khrapko⁶, Robert W. Taylor², Thomas B. L. Kirkwood⁷, John C. Mathers^{1,5}, Douglass M. Turnbull^{1,2}

1 Newcastle University Centre for Brain Ageing and Vitality, Institute for Ageing and Health, Newcastle University, Newcastle upon Tyne, United Kingdom, 2 Wellcome Trust Centre for Mitochondrial Research, Institute for Ageing and Health, Newcastle University, Newcastle upon Tyne, United Kingdom, 3 Institute of Genetic Medicine, Newcastle University, Newcastle upon Tyne, United Kingdom, 4 Centre for Human Metabolomics, North-West University, Potchefstroom, South Africa, 5 Human Nutrition Research Centre, Institute for Ageing and Health, Newcastle University, Newcastle upon Tyne, United Kingdom, 6 Beth Israel Deaconess Medical Center, Boston, Massachusetts, United States of America, 7 Institute for Ageing and Health, Newcastle University, Campus for Ageing and Vitality, Newcastle upon Tyne, United Kingdom

Abstract

Age-related decline in the integrity of mitochondria is an important contributor to the human ageing process. In a number of ageing stem cell populations, this decline in mitochondrial function is due to clonal expansion of individual mitochondrial DNA (mtDNA) point mutations within single cells. However the dynamics of this process and when these mtDNA mutations occur initially are poorly understood. Using human colorectal epithelium as an exemplar tissue with a well-defined stem cell population, we analysed samples from 207 healthy participants aged 17–78 years using a combination of techniques (Random Mutation Capture, Next Generation Sequencing and mitochondrial enzyme histochemistry), and show that: 1) non-pathogenic mtDNA mutations are present from early embryogenesis or may be transmitted through the germline, whereas pathogenic mtDNA mutations are present from early adulthood (<20 years of age), at both low levels and as clonal expansions, 3) low level mtDNA mutation frequency does not change significantly with age, suggesting that mtDNA mutation rate does not increase significantly with age, and 4) clonally expanded mtDNA mutations increase dramatically with age. These data confirm that clonal expansion of mtDNA mutations, some of which are generated very early in life, is the major driving force behind the mitochondrial dysfunction associated with ageing of the human colorectal epithelium.

Citation: Greaves LC, Nooteboom M, Elson JL, Tuppen HAL, Taylor GA, et al. (2014) Clonal Expansion of Early to Mid-Life Mitochondrial DNA Point Mutations Drives Mitochondrial Dysfunction during Human Ageing. PLoS Genet 10(9): e1004620. doi:10.1371/journal.pgen.1004620

Editor: Nils-Göran Larsson, Max Planck Institute for Biology of Ageing, Germany

Received February 26, 2014; Accepted July 21, 2014; Published September 18, 2014

Copyright: © 2014 Greaves et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: This research was supported by the Food Standards Agency (N12015), NIHR Newcastle Biomedical Research Centre based at Newcastle upon Tyne Hospitals National Health Service Foundation Trust and Newcastle University, the Wellcome Trust Centre for Mitochondrial Research (096919Z/11/Z), the Medical Research Council (UK) Centre for Translational Muscle Disease research (G0601943) and the UK NHS Highly Specialised "Rare Mitochondrial Disorders of Adults and Children" Service, the Newcastle University Centre for Brain Ageing and Vitality supported by the BBSRC, EPSRC, ESRC, and MRC as part of the cross-council Lifelong Health and Wellbeing Initiative (G0700718), the Ellison Medical Foundation and the NIA(R01AG19787). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that there are no competing interests.

* Email: laura.greaves@ncl.ac.uk

Introduction

Mutations of mitochondrial DNA (mtDNA) have been implicated in the ageing process [1]. As humans age, multiple different mutations arise somatically in individual cells and some of these expand clonally to high levels over time, resulting in focal respiratory chain deficiencies [2–6]. To date there is poor understanding of the dynamics of mutation accumulation during ageing and of when in the life-course the majority of these clonally expanded somatic mtDNA mutations initially occur [7]. For example, there have been suggestions that the underlying mechanism involves either an accelerating mtDNA mutation rate over time [8], or clonal expansion of mtDNA mutations which have occurred in early life [7]. Resolving these questions about the dynamics of mtDNA point mutations is important because of their accumulation in human stem cell populations with age, which results in respiratory chain dysfunction [5,9–14] and consequent reductions in cell function [15]. Age-related dysfunction of somatic stem cells has been proposed to lead to the decreased ability of tissues to regenerate [16]. Mice with increased mtDNA mutagenesis (mutator mice) caused by a defect in the proof-reading ability of the mitochondrial DNA polymerase gamma also show a premature ageing phenotype [17,18] which has been attributed largely to somatic stem cell dysfunction resulting from high mtDNA point mutation loads [19–22]. In mutator mice the majority of the mutational burden leading to a cellular phenotype occurs during embryogenesis [19]. However, in human tissues there have been no comprehensive

Author Summary

Mitochondrial DNA (mtDNA) mutations have been shown to accumulate with age in a number of human stem cell populations and cause mitochondrial dysfunction within individual cells resulting in a cellular energy deficit. The dynamics by which mtDNA mutations occur and accumulate within individual cells (known as clonal expansion) is poorly understood. In particular we do not know when in the life-course these mtDNA mutations occur. Here we have measured mtDNA mutation frequency using three different techniques; Random Mutation Capture, which measures low level mutation frequency as an indirect measure of mutation rate, Next Generation Sequencing, which measures clonally expanded mtDNA mutation frequency, and mitochondrial enzyme histochemistry as a marker of clonally expanded mtDNA mutations, on colorectal mucosal biopsies obtained from 207 healthy participants aged 17-78 years. We show that, by 17 years of age, there is a substantial mtDNA point mutation burden and that clonal expansion of early to mid-life mtDNA mutations is likely to be the cause of mitochondrial dysfunction associated with ageing in the human colon.

studies examining mtDNA point mutation occurrence and accumulation over the life-course.

Here, we apply different validated techniques to investigate the frequency of both low level (as an indirect measure of mutation rate) and clonally expanded mtDNA mutations, using the human colonic epithelium as an exemplar tissue with a well-characterised stem cell population. We found no evidence of a significant increase in the frequency of low level mtDNA mutations with age, but there was a significant increase in the frequency of clonally expanded mtDNA mutations with age. We provide robust evidence that mtDNA mutations occur early in life and that a substantial mtDNA point mutation burden exists within the human colorectal epithelium before the age of 20.

Results

No significant increase in low level mtDNA mutation frequency with age

Colorectal biopsies from 207 subjects aged 17-78 years with no evidence of bowel pathology at endoscopy were collected. Low level mtDNA mutation frequencies were quantified in these biopsy samples using a highly sensitive Random Mutation Capture (RMC) assay [23,24] (Figure 1A). Approximately 200 million base pairs of mtDNA sequence were screened and a total of 803 mutations were detected. All mutations and full details of the number of bases investigated per individual are shown in Table S1. Examination of the types of mutational events detected by RMC in our cohort showed that 60% of all mtDNA mutations were G>A or C>T transitions, 24% were T>C or A>G transitions, and the remainder were transversions and small insertions and deletions. We also noted an uneven distribution of mutations across the four base pair TCGA TaqIa site; 63% of changes were at the third base pair, with the remaining 37% spread fairly evenly across the other 3 bases. There was no significant correlation between low level mtDNA mutation frequency and age (Pearson correlation = 0.127 (p = 0.07, Figures 1B and 1C)). The RMC assay can be intrinsically noisy due to random sampling statistics [25], with inter-individual variation often observed in studies of ageing populations [26]. Therefore to maximise our chance of detecting a relationship between low level mtDNA mutation frequency and age, we pooled the data by decade of participant age. Although there was a modest increase in low-level mutation frequency with age (Figure 1D), this was not statistically significant (p = 0.343, One Way ANOVA). A Tukey post-hoc comparison revealed no significant differences even for the comparison between the first and the last decade of participant age. In a number of subjects we were not able to detect any mtDNA mutations in the base pairs screened. To ensure that these zero values were not having a significant effect on the data we reran the analyses excluding the zero values. There was still no significant association with age in either the individual data points (p = 0.07, Pearson correlation = 0.136) or the grouped data (p = 0.46, p)One Way ANOVA) (Figure S1). These data highlight that even the youngest person studied (aged 17 years) had an appreciable mtDNA mutation frequency of 2.5 mutations per 10^6 base pairs. Collectively these analyses demonstrate no significant increase in low level mtDNA mutation frequency with age. There was no significant difference in the frequency of synonymous and non-synonymous mtDNA mutations in older (>46 years) vs younger participants (<46 vears) (p = 0.665, Fisher's exact test). This confirms that there are no selective pressures acting on mtDNA point mutation occurrence with age and that mtDNA mutations present from early adulthood in the human colon could be pathogenic later in life if they were to clonally expand to high levels over time.

Significant increase in the frequency of clonal expansions and low level heteroplasmic mtDNA mutations with age

Due to the small size of the region of the genome under investigation by the RMC assay, and the relative rarity of clonally expanded mutations across the entire mtDNA molecule, the RMC will not detect most of the clonally expanded mtDNA mutations. Indeed, in our RMC mutation dataset there were no mutations expanded to more than the equivalent of 1/100 of a crypt size. Therefore, we employed a next generation sequencing (NGS) approach to examine all mtDNA sites and gain information about the age dependent dynamics of clonally expanded mutations. Whole mtDNA Ion Torrent NGS was carried out on DNA extracted from the biopsies from a representative subset of the youngest (≤ 26 years of age n = 8) and oldest participants (≥ 70 years of age, n = 8) investigated by RMC. Our stringent quality control criteria (see Materials and Methods) set the heteroplasmy threshold for calling mtDNA mutations at 0.8%. Based on this figure and the average number of crypts per biopsy (~ 200), our NGS assay could detect homoplasmic clonal expansions within individual crypts or clusters of clonal crypts [14], as well as low levels of heteroplasmic mtDNA mutations (>0.8%) present throughout the whole tissue [27,28]. Figure 2 details the various techniques employed throughout the study and the limits of detection for each technique.

We sequenced a total of 556 Mb of mtDNA (average of 35 Mb per subject) and detected 109 mutations present at >0.8% heteroplasmy. All detected mtDNA variants are detailed in Table S2. All participants showed some variants at >0.8% heteroplasmy, but the frequency of mtDNA mutations was more than 8-fold higher in the older than in the younger group (Figure 3A, p = 0.036, unpaired t-test). There was no significant difference in the types (transitions/transversions) of mutations observed between the two age groups (Figure 3B, p = 1.00, Fisher's exact test), with single nucleotide transitions being by far the major mutation type.

A recent study has shown that recurrent tissue-specific mtDNA mutations are present in unrelated individuals [29]. We investigated our NGS data to determine whether recurrent mtDNA mutations were also present within our dataset. In our colon data we detected eight unique mtDNA mutations that were present in



Figure 1. Analysis of mitochondrial DNA point mutation frequency with age by Random Mutation Capture (RMC). (A) Schematic diagram describing the RMC methodology. (i) Schematic diagram of the structure of the human colorectal crypt. (ii) Schematic diagram showing mtDNA isolation. Colonoscopic biopsies are homogenized and the mitochondrial fraction isolated by differential centrifugation. MtDNA is then prepared by phenol/chloroform extraction and quantified using real-time PCR (standard curve method). (iii) MtDNA is digested for 10 hours with *Taq1a*. PCR is then carried out over the restriction site. Only molecules with mutations in the restriction site are able to successfully amplify. (iv) Agarose gel showing PCR products from a typical RMC run. Each reaction contained ~10000 target base pairs. 488 base pair bands show amplified mutated molecules (wells 4,6,13,16 and 20). The wild-type control well (Wt) shows complete digestion of wild-type DNA following PCR. (v) Example electropherograms showing mutations (asterisks) within the restriction site (highlighted in blue). (B) Frequency of all RMC detected mtDNA mutations in human colorectal mucosa (n = 207). There was no correlation between mtDNA mutation frequency and age (Pearson correlation = 0.127 (P = 0.07)). (C) Data from (B) presented on a log 10 scale to show the spread of the data. Note that the zero values cannot be displayed in this way therefore n = 175. (D) Frequency of all mtDNA mutations detected in human colonic mucosa, grouped by decade. Subjects were grouped as follows, 17-26 (n = 12), 27-36 (n = 19), 37-46 (n = 58), 47-56 (n = 51), 57-66 (n = 43), 67-77 (n = 23). There were no significant differences between any of the groups (P = 0.343, One Way ANOVA).

doi:10.1371/journal.pgen.1004620.g001

two individuals; however unlike the data from Samuels et al, they were not restricted to the non-coding regions, but appeared to be located randomly; three were in the non-coding region, three were in protein encoding genes and two were in RNA genes. Five of these were previously reported polymorphic variants [30], the remaining three were previously unreported. There is no evidence of contamination as they do not fit a haplogroup specific pattern with multiple markers seen in individual subjects [31], they were present in random pairs of subjects, and they were not observed in the yeast plasmid control.

Exclusion of low level inherited heteroplasmic mtDNA mutations

It has recently been shown that mtDNA mutations in adult tissues can originate in embryonic development or even in the germline [27,28], inferring that some of the mutations that we detected by NGS could have occurred during this period. To investigate this possibility, matched buccal scrape samples were collected at the same time as the colorectal mucosal biopsies from the same 16 participants from whom we had carried out NGS (data in Figure 3), and NGS was performed on DNA from the buccal cells. Identification of the same mtDNA mutations in two different tissues would support the hypothesis that such mutations occurred prior to tissue differentiation during embryogenesis. It should be noted that buccal and colonic epithelial cells both arise from the endoderm with the fore and hind gut becoming separate tissues by weeks 3-4 of gestation [32]. In DNA from the 16 participants investigated, we detected a total of 16 mtDNA mutations that were present at low levels of heteroplasmy in both tissues (Table S2 and Table S3) and these occurred in 10 participants. Five of these people were in the >70 year age group,

TECHNIQUE	TYPES OF MTDNA MUTATIONS DETECTED
Next Generation Sequencing	Inherited or somatic germline heteroplasmy at the tissue level (>0.8%) Homoplasmic intracellular clonal expansions at the individual crypt level
COX/SDH Histochemistry	High level intracellular somatic clonal expansions (~>60% - 100%)
Single Crypt Sequencing	Mid-high (25%-100%) level intracellular somatic clonal expansions
Random Mutation Capture	Very low level mtDNA mutations (down to 0.0001% at the tissue level) Clonal expansions (intracellular or germline) within the 4 base pair RMC site

Figure 2. Schematic diagram showing the detection limits of each of the techniques utilised in this study. doi:10.1371/journal.pgen.1004620.g002

five were in the <26 year age group. There was no significant difference in the frequency of germline or embryological mtDNA mutations between the <26 year and >70 year age group (p = 0.176, unpaired t-test) confirming that there was no age-effect and that these mtDNA mutations were most likely of germline or embryological origin.

The frequency of somatic mtDNA mutations in the colon samples was then analysed by subtracting the germline or embryological mtDNA mutations from the total mtDNA mutation frequency. This revealed a significant 10-fold increase in the frequency of clonally expanded somatic mtDNA mutations in those aged >70 years compared with <26 years (Figure 4A, p = 0.035, unpaired t-test). From here on the mtDNA mutations detected in colorectal epithelium only will be referred to as somatic mtDNA mutations and those present in both buccal and colorectal epithelium as germline or early embryological mtDNA mutations. The pattern of somatic mtDNA mutations detected in the buccal epithelium was similar to those in the colonic epithelium. The mtDNA mutations detected were base transitions and were randomly located throughout the genome. We did observe a higher number of somatic mtDNA mutations in the colonic epithelial samples compared to the buccal samples. We have previously shown that there are tissue specific differences in the frequency of clonally expanded mtDNA mutations, with the colon

being one of the most highly affected [10] and believe that this could explain these differences.

Next we compared the ratio of synonymous or polymorphic protein encoding mtDNA mutations to non-synonymous mtDNA mutations in the somatic and germline or early embryological data sets to see if there were any differences between the two, and therefore any evidence for purifying selection. There was a significantly higher proportion of non-synonymous mtDNA mutations in the somatic data set compared with the germline or early embryological data set (Fisher's exact test with, p = 0.041, Figure 4B); in fact only one of the germline or early embryological mtDNA mutations was non-synonymous and therefore potentially pathogenic. These data suggest that the mtDNA mutations likely to contribute to the mitochondrial ageing phenotype begin to occur sometime after 3-4 weeks gestation (1-2 weeks postconception), which coincides with the resumption of mtDNA replication which is thought to occur post-embryo implantation [33].

A significant increase in the frequency of respiratory chain deficient colonic crypts with age correlates with the frequency of mutations detected by NGS

We have previously shown that the frequency of crypts deficient in cytochrome c oxidase activity (complex IV of the respiratory



Figure 3. Mitochondrial DNA mutations in human colorectal epithelium of subjects below 26 years of age (n = 8) and over 70 years of age (n = 8) measured by Next Generation Sequencing (NGS). (A) MtDNA mutation frequency. There was a significantly higher mutation frequency in the subjects >70 years (p = 0.0361, unpaired t-test). (B) Types of mutations detected by NGS frequency. There was no significant difference in the types of mutations detected in subjects <26 and >70 years of age (p = 1.00, Fisher's exact test). doi:10.1371/journal.pgen.1004620.g003

chain) increases with age in the apparently normal mucosa taken from patients with a colorectal tumour [5]. We also demonstrated that in the vast majority of these crypts there is an intracellular clonally expanded mtDNA point mutation [5,34]. The present study provided an opportunity to determine whether there was a similar age-related increase in the frequency of crypts deficient in cytochrome c oxidase activity in healthy participants in whom there was no evidence of mucosal dysplasia. In addition, as cytochrome c oxidase (COX)/succinate dehydrogenase (SDH) histochemistry is an excellent surrogate marker for mid-high level intracellular clonally expanded mtDNA point mutations, which both RMC and NGS are quite likely to miss, this assay gives an indication of the frequency of such mutations (Figure 2). Colorectal mucosal biopsies collected by endoscopy from the same 207 subjects investigated by the RMC assay were subjected to sequential COX/SDH histochemistry (Figure 5A) and the percentage of COX deficient colonic crypts calculated. As expected, there was a significant increase in the percentage of COX deficient crypts in individuals with age (Figure 5B, Pearson correlation 0.603 (p<0.001)). The somatic mtDNA mutations detected using the Ion Torrent NGS platform could be clonally expanded mtDNA mutations in individual colonic crypts or low level clonally expanded mtDNA mutations present throughout the whole tissue. Therefore we investigated a possible correlation between the percentage of COX deficient crypts (known to be a good marker of clonally expanded mtDNA point mutations [5]) and the somatic mtDNA mutation frequency measured by NGS. This showed that there was a significant correlation between mtDNA mutation frequency as measured by NGS and COX crypts (Figure 5C, Pearson correlation = 0.511deficient (p = 0.043)). When we compared the germline mtDNA mutation frequency with the percentage of COX deficient crypts, there was no significant correlation (Figure 5D, Pearson correlation = 0.369, p = 0.176). There was no significant correlation between the



Figure 4. Exclusion of mitochondrial DNA (mtDNA) mutations occurring in the germline or in early embryogenesis. (A) Somatic mtDNA frequency (mtDNA mutations present in colon only) measured by NGS. There was a significantly higher mutation frequency in the subjects > 70 years (P = 0.0351, unpaired t-test). (B) Percentage of synonymous/polymorphic variants and non-synonymous mtDNA mutations which were of germline or early embryological origin compared to those which were somatic in adults <26 years of age and over 70 years of age. There was a significantly lower frequency of non-synonymous mtDNA mutations in the somatic mtDNA mutation groups compared to the germline or early embryological mtDNA mutation group (p = 0.041 Fisher's, exact test). doi:10.1371/journal.pgen.1004620.g004



Figure 5. Quantification of mitochondrial dysfunction in human colonic crypts by cytochrome *c* oxidase (COX)/succinate dehydrogenase (SDH) histochemistry. (A) COX/SDH sequential histochemistry on a transverse section of human colorectal mucosa. The crypts which stain brown have functional COX and SDH activity; those which are blue have lost COX activity but retain SDH activity. (*i*) shows an image from a subject aged 25 in whom no COX deficient crypts were detected. (*ii*) shows an image from a subject aged 72 in whom 28% of crypts were COX deficient (**B**) COX activity was measured in colorectal biopsies from 207 subjects. A crypt was defined as deficient if more than 50% of cells had lost COX activity. There was a significant correlation between age and the percentage of COX deficient crypts. (Pearson correlation = 0.603, p<0.0001). (**C**) Correlation analysis between somatic mtDNA mutation frequency determined by NGS and percentage COX deficient crypts showed a significant correlation = 0.511, p = 0.043). (**D**) Correlation analysis between germline mtDNA mutation frequency determined by NGS and percentage COX deficient crypts showed no significant correlation = 0.369, p = 0.176). (**E**) Correlation analysis between the frequency of random mtDNA mutations by RMC and percentage COX deficient correlation (Pearson correlation = 0.007, p = 0.918). (**F**) Correlation analysis between the frequency of random mtDNA mutations by RMC and percentage showed no significant correlation (Pearson correlation = 0.145, p = 0.381). doi:10.1371/journal.pgen.1004620.g005

mtDNA mutation frequency detected by RMC and COX deficiency (Figure 5E, Pearson correlation = 0.007, p = 0.918), suggesting that the majority of pathogenic mtDNA mutations detected by NGS are somatic clonally expanded variants. In addition when we compared the RMC and NGS data from the same subjects side by side, there was no significant correlation confirming that the two assays were measuring different classes of mtDNA mutations i.e. low level vs clonally expanded (Figure 5F, Pearson correlation = 0.381, p = 0.145, Figure 2).

Discussion

In this study we have examined the timing of occurrence and frequency of mtDNA mutations during ageing in human colorectal epithelium. We have employed a range of methodologies to provide accurate assessment of low level mtDNA mutation frequency, germline mtDNA heteroplasmy, and both high and low level intracellular clonal expansion. For low level mtDNA mutation frequency, we have shown previously that the most accurate method to use is the RMC assay [25], which provides the best available measure of mtDNA mutation rate [23]. An advantage of this assay is that it is not affected by false positive mutational calls caused by either PCR or sequencing errors; however a limitation is that, because only a small mtDNA domain is interrogated, it will miss the majority if not all intracellular clonally expanded mtDNA mutations. The NGS approach we have used is, in our hands, sensitive down to 0.8% heteroplasmy which correlates to either low-level germline mtDNA heteroplasmy or homoplasmic intracellular mtDNA clonal expansions in one or more crypts [14] (based on there being ~ 200 crypts in a colonoscopic biopsy sample). Sequencing of individual laser microdissected crypts, which we have done extensively in our previous work [5,34], is required to detect the remaining mtDNA mutations, i.e. mid-high level intracellular clonal expansions; these can also be detected using COX/SDH histochemistry as a surrogate marker [5,14,34,35] as we have done here (Figure 2). There is the possibility that there are effectors of COX deficiency other than clonally expanded mtDNA point mutations, such as nuclear DNA mutations or changes in global gene expression; however we have previously found evidence of a clonally expanded mtDNA point mutation in >75% of crypts sequenced [5,14,34,35] and therefore believe that this is an excellent surrogate marker. We have not previously detected any largescale mtDNA deletions in individual COX deficient crypts [5].

The RMC assay revealed that there was no significant increase in low level mtDNA mutation frequency with age. This is supported by a published study which used RMC to measure mtDNA mutation frequency in colonic epithelium from a much smaller cohort of individuals covering a narrower age range ($\sim 50-$ 90 years of age (n = 20) [23]. An important result of our study is a demonstration that the lack of an increase in low level RMCdetected mtDNA mutations (i.e. no increase in mtDNA mutation rate) does not mean that there is no increase in total mtDNA mutation load with age. Instead, we show that clonally expanded mutations, as measured by NGS, increase very dramatically. This is the first demonstration that mtDNA mutation rate and clonal expansion may follow very different age dynamics. This is in direct contradiction of the mitochondrial vicious cycle hypothesis of ageing [36] which suggests that mtDNA mutations occur during ageing leading to dysfunctional proteins in the oxidative phosphorylation system, precipitating increased mutation i.e. an accelerating mtDNA mutation rate over time. Our results also demonstrate that caution should be exercised in interpreting mutation analysis results, which may be limited to only a portion of mutations depending on the technique used.

Together the NGS and RMC datasets suggest that mtDNA mutation rate does not change significantly with age, but that clonal expansion of mtDNA mutations occurs over time. Mathematical modelling studies have suggested that clonal expansion of mtDNA mutations within an individual cell is likely to be due to random genetic drift and predict that it can take at least 20 years for an mtDNA mutation to clonally expand to high levels sufficient to cause COX deficiency [37-39]. Indeed, the youngest participant in this cohort in whom COX deficiency was detected was 21 years of age, therefore the initial mutational event(s) in this case must have occurred very early in life. Interestingly, the mtDNA mutation frequency data by NGS from our human samples were different to those obtained from a similar NGS study carried out in mice in which no significant increase in mtDNA mutation frequency with age was noted [40]. Previously we examined colonic epithelial tissue from a similar ageing mouse colony and showed that clonal expansion of mtDNA mutations was a very rare event in these animals compared with aged humans [41]. This may explain the species differences in these data, consistent with modelling studies that emphasise the difficulty of generating clonal expansion through random drift in short-lived animals [39].

Whilst our data imply that mutations of very early origin contribute to mitochondrial dysfunction in old age, it does not mean that mutations occurring in adult life play no role. In fact, the number of different clonally expanded mutations per sample detected by NGS robustly increased with age (3-fold from <26 years to >70 years group, p = 0.001). Such an increase in mtDNA point mutation diversity can be explained with a scenario whereby *de novo* mtDNA point mutations occurring during adult life, perhaps up to middle-age, are able to clonally expand and join the set of expanded mutations detected in old age. However mtDNA mutations which occur late in life will not have time to expand to high levels.

Analysis of the RMC data showed that even in the youngest participants, we observed a substantial load of mtDNA mutations in the colonic epithelium. NGS analysis of a subset of our youngest participants, all of whom were <26 years of age (n = 8), confirmed the RMC data; young adults have a significant mtDNA point mutation load. This has been previously shown to be the case in DNA extracted from young brain [42,43], where both point mutations and mtDNA deletions have been detected. Our RMC data have now shown that this is also the case for a mitotic tissue, the colonic epithelium. In addition we have clearly demonstrated that the same type of mtDNA mutations (point mutations which are predominantly transitions) are present in young individuals as those detected in our previous studies of clonally expanded mtDNA mutations from aged respiratory chain deficient individual crypts [34], i.e. the seed mutations for clonal expansion can be laid down at an early age. Although this has been predicted by modelling simulations [37,38] this is the first experimental evidence to show this definitively.

Our NGS analysis has shown that low level heteroplasmic mutations are present in multiple tissues from the same individual. This supports previous studies showing that mtDNA mutations in adult tissues can originate in germline or very early development [27,28]. Indeed, our data from the colonic and buccal epithelium show that mtDNA mutations present in both tissues must have occurred prior to the fore and hind guts becoming separate which is thought to occur 1-2 weeks post-conception [32]. It is possible that due to our conservative cut off of 0.8% heteroplasmy, there may be additional germline mtDNA mutations which have undergone less drastic clonal expansion in one of the two tissues studied than the other, and therefore are below the threshold of detection; this is one of the limitations of the available technology. These data do show that 95% of the heteroplasmic mutations detected in both tissues were non-pathogenic polymorphic variants, thus suggesting that pathogenic mtDNA mutations which occur in the germline or early development are selected against, and these non-pathogenic mtDNA mutations may make little contribution to the ageing phenotype. This demonstrates purifying selection in the human germline. Previous studies have shown this in mice by looking at transmission of mtDNA mutations through multiple generations [44,45]; here we show purifying selection in humans by comparing the germline and somatic mtDNA mutations in different tissues from the same subjects. Recent evidence from the mouse has suggested that transmitted germline mtDNA mutations can cause premature ageing, perhaps by clonal expansion of these germline mtDNA mutations over time [46]. In our dataset 95% of the germline mtDNA mutations are benign and are unlikely to cause mitochondrial dysfunction and premature ageing, suggesting that there are differences in the dynamics of mtDNA transmission between these mutation prone mice and humans.

The somatic mtDNA mutations detected by NGS in the colonic epithelium only, are a combination of benign synonymous and polymorphic variants and non-synonymous potentially pathogenic variants, which we believe may begin to occur when mtDNA replication is re-initiated after the embryo has implanted into the uterine wall [33]. There was a significantly higher frequency of non-synonymous coding region somatic mtDNA mutations compared to the germline or early embryological mtDNA mutations which is evidence in support of the hypothesis that the somatic mutations occurred beyond any selective checkpoints, before expanding clonally to detectable levels.

The observations in this study are in agreement with evidence from epidemiological studies which suggest that damage arising early in human life can be an important modulator of outcomes in later life [47,48]. Due to the time taken for clonal expansion of mtDNA mutations to occur in human cells, we hypothesise that late life *de novo* mtDNA mutational events make negligible contribution to the ageing phenotype and that early to mid-life mtDNA mutations are likely to be much more important.

Materials and Methods

Participants

Colorectal mucosal samples were collected from the same anatomical site (10 cm from the anal verge) from participants (n = 207, age range 17–78 years) undergoing colonoscopy for disturbed bowel function in whom no evidence of bowel disease was identified (BORICC 1 Study). Buccal epithelial scrapes were also collected concurrently from these subjects. The following subjects were also used in our previous work: BCC010, BCC011, BCC017, BCC022, BCC028, BCC085, BCC087, BCC088 [25].

Ethics statement

Ethical approval was obtained from the Northumbria NHS Trust Local Research Ethics Committee (Project reference NLREC2/2001). All participants were fully informed and written consent obtained from them.

Random Mutation Capture (RMC)

RMC was carried out essentially as previously described [25]. Briefly, mtDNA was extracted from colorectal mucosal biopsies and drop-dialysed using membrane filters (0.025 μ m, Millipore) to extract any excess salts. One microlitre of mtDNA was digested with 100 U of TaqIa (New England Biolabs) for 10 hours with the addition of 100 U every hour. MtDNA copy number was quantified by SYBR Green real-time PCR (Roche) targeting a template outside of a $TaqI\alpha$ restriction site in MTND5 (primers L12473-L12492 and H12573-H12554) Absolute quantification was carried out using the standard curve method. PCR was then carried out across a $TaqI\alpha$ restriction site within the *MTCOI* gene (bp 6562-6565, primers L4636-L6455 and H6851-H6870). An average of 2500 copies of the target sequence (a total of 10000 target bases) was added to each PCR reaction. Following PCR each product was digested with 50 units of TaqIa for 1 hour at 65°C, followed by 10 minutes at 80°C to inactivate the enzyme. Products were then subjected to electrophoresis through a 1.5% agarose gel for 1 hour at 200 V. All full length (488 bp) products were excised from the gel using a QIAquick Gel Extraction kit (Oiagen). These products were sequenced using ABI BigDve chemistries per standard manufacturer's protocols and analysed on an ABI3100 Genetic Analyser (Applied Biosystems). Sequences obtained were compared with the revised Cambridge Reference Sequence (GenBank accession number: NC_012920.1) using SeqScape software (Applied Biosystems). Mutation load was calculated by dividing the number of confirmed mutants by the total number of base pairs investigated.

Validation of the RMC assay

To investigate the sensitivity and specificity of the RMC assay in our hands we generated a PCR product which contained a mtDNA mutation in the RMC site and one which was wild-type in the RMC site. A pCR-scriptTM Amp SK(+) cloning Kit (Stratagene) was used to clone the products following the manufacturer's protocol. Recombinant plasmids were identified by blue–white colour selection and the cultures grown up using a Qiaquick miniprep kit (Qiagen). The DNA was then extracted and quantified and the wild-type and mutant PCR products mixed at concentrations ranging from 100% wild-type to 100% mutant. The RMC assay was then carried out as above and the observed mutant fractions calculated and compared to the expected fractions (Table S4). There was no difference between observed and expected fractions, confirming the RMC assay to be both highly sensitive and specific.

Next generation sequencing

Next generation sequencing (NGS) was carried out using an Ion Torrent Personal Genome Machine (Life Technologies, Paisley, UK) on whole mtDNA from the same colonic biopsies investigated by RMC and from buccal epithelia from the 8 youngest (<26 years) and 8 oldest (>70 years) subjects. To exclude the possibility of nuclear pseudogene amplification, extracted DNA was amplified in two overlapping 9 kb fragments using primers L2091– L2111 and H10649-H10629 (primer set 1), and L10085–L10104 and H2644-2625 (primer set 2), the specificity of which was established after observing no amplification from Rho Zero cells, cells depleted of their mtDNA by ethidium bromide treatment.

Long-range PCR amplicons were quantified on an Agilent 2100 Bioanalyzer with an Agilent DNA 12,000 kit (Agilent Technologies, Stockport, UK). Overlapping PCR fragments for each sample were combined in equimolar concentrations. Pooled amplicons (100 ng) were then fragmented, barcoded, size-selected and amplified using the IonXpress Plus Fragment Library kit, Ion Xpress Barcode Adapters and E-Gel SizeSelect 2% agarose gels (Life Technologies), according to the manufacturer's recommendations. Barcoded libraries were quantified with an Agilent Bioanalyzer DNA High Sensitivity kit then pooled (n = 16) in equimolar concentrations and diluted to 26 pM, prior to clonal amplification onto Ion Sphere Particles using the Ion OneTouch 1 System and the Ion OneTouch 200 Template kit v2 (Life Technologies), as per the manufacturer's instructions. Coated spheres were enriched on the Ion Torrent ES (Life Technologies) before loading onto Ion 318 sequencing chips (Life Technologies). Next-generation semiconductor sequencing was performed on an Ion Torrent Personal Genome Machine (Life Technologies). Fastq data files downloaded from the Torrent Server (version 3.6.2, Life Technologies) were analysed using NextGENe software (v2.3.0; SoftGenetics, State College, PA, USA).

Quality control criteria for NGS

The background noise on the Ion Torrent PGM platform was quantified by extraction of DNA from a yeast clone which had been transfected with a plasmid (pRShmt) containing the entire human mtDNA [49] (kindly donated by Dr Brian Bigger (University of Manchester, UK)) which was then subjected to an identical PCR amplification and NGS protocol as the colonic biopsy and buccal scrape samples. On this basis any low level mtDNA variants detected in the cloned mtDNA are likely to be technical artefacts arising from the PCR and sequencing process and we could quantify the level of background noise and exclude this from the sample analysis. In addition the Ion Torrent platform may be prone to base calling errors in polynucleotide tracts, most often calling them as small insertions or deletions; therefore we restricted the analysis to base-pair substitutions only. Further quality control steps taken were; (1) only base substitutions with a quality score >20 were included in order to be confident that the calls were genuine, (2) observed variants had to be present in both forward and reverse reads at comparable frequencies with a 3-fold difference permitted to allow for the effects of a binomial sampling distribution at very low variant levels [28], (3) at least 3 reads were required for each variant, with a minimum total coverage of 600 reads per site. There was an unstable tract between base pairs 3902 and 3908 which repeatedly showed heteroplasmy levels between 1% and 5% in all of the samples and the plasmid control, as did a recognised variant at base pair 750. These mtDNA variants were deemed artefactual and removed from the analysis. Using these stringent criteria, there were no variants present at >0.65% in the cloned DNA template (Table S5). We took a conservative approach and only recorded mutations present at >0.8%. This approach ensured that any variants detected in the samples at levels of >0.8% are likely to be generated in vivo and be of biological origin. The published base-substitution error rate for mtDNA on the Ion Torrent PGM is 0.12% [50].

References

- Greaves LC, Turnbull DM (2009) Mitochondrial DNA mutations and ageing. Biochim Biophys Acta 1790: 1015–1020.
- Fellous TG, Islam S, Tadrous PJ, Elia G, Kocher HM, et al. (2009) Locating the stem cell niche and tracing hepatocyte lineages in human liver. Hepatology 49: 1655–1663.
- Muller-Hocker J (1990) Cytochrome c oxidase deficient fibres in the limb muscle and diaphragm of man without muscular disease: an age-related alteration. J Neurol Sci 100: 14–21.
- Muller-Hocker J, Schneiderbanger K, Stefani FH, Kadenbach B (1992) Progressive loss of cytochrome c oxidase in the human extraocular muscles in ageing-a cytochemical-immunohistochemical study. Mutat Res 275: 115–124.
- Taylor RW, Barron MJ, Borthwick GM, Gospel A, Chinnery PF, et al. (2003) Mitochondrial DNA mutations in human colonic crypt stem cells. J Clin Invest 112: 1351–1360.

Cyctochrome c oxidase/succinate dehydrogenase (COX/ SDH) histochemistry

Colon samples were mounted for sectioning and frozen in isopentane previously cooled to -190° C in liquid nitrogen. Cryostat sections (12 µm) were cut onto glass slides and incubated in COX medium (100 µM cytochrome *c*, 4 mM diaminobenzidine tetrahydrochloride and 20 µg.ml⁻¹ catalase in 0.2 M phosphate buffer pH 7.0) at 37°C for 50 minutes. Sections were washed in phosphate buffered saline, pH 7.4 (3×5 minutes) and incubated in SDH medium (130 mM sodium succinate, 200 µM phenazine methosulphate, 1 mM sodium azide, 1.5 mM nitroblue tetrazolium in 0.2 M phosphate buffer pH 7.0) at 37°C for 45 minutes. Finally, sections were washed in phosphate buffered saline, pH 7.4 (3×5 minutes), dehydrated in a graded ethanol series (70%, 95%, 2×100%), cleared in Histoclear (National Diagnostics, Atlanta, USA) and mounted in DPX.

Supporting Information

Figure S1 RMC analysis with zero values removed. (PDF)

Table S1Random mutation capture raw data.(PDF)

 Table S2
 Mitochondrial DNA (mtDNA) mutations detected by

 Ion
 Torrent
 Next
 Generation
 Sequencing in human colonic

 epithelium.
 (PDF)

Table S3 Mitochondrial DNA (mtDNA) mutations detected byIon Torrent Next Generation Sequencing in human buccalepithelium.

(PDF)

Table S4Validation of the sensitivity and specificity of the RMC assay.

(PDF)

Table S5 MtDNA variants detected in the cloned mtDNA plasmid control.

(PDF)

Acknowledgments

We thank DM Bradburn (Wansbeck General Hospital) for his assistance in recruiting study participants and Siegfried Hekimi for helpful discussions.

Author Contributions

Conceived and designed the experiments: LCG DMT JCM TBLK. Performed the experiments: LCG MN HALT GAT. Analyzed the data: LCG JLE KK RWT. Contributed reagents/materials/analysis tools: DMC RPA. Wrote the paper: LCG JLE DMT.

- Yu-Wai-Man P, Lai-Cheong J, Borthwick GM, He L, Taylor GA, et al. (2010) Somatic mitochondrial DNA deletions accumulate to high levels in aging human extraocular muscles. Invest Ophthalmol Vis Sci 51: 3347–3353.
- Khrapko K (2011) The timing of mitochondrial DNA mutations in aging. Nat Genet 43: 726–727.
- Linnane AW, Marzuki S, Ozawa T, Tanaka M (1989) Mitochondrial DNA mutations as an important contributor to ageing and degenerative diseases. Lancet 1: 642–645.
- Blackwood JK, Williamson SC, Greaves LC, Wilson L, Rigas AC, et al. (2011) In situ lineage tracking of human prostatic epithelial stem cell fate reveals a common clonal origin for basal and luminal cells. J Pathol 225: 181–188.
- Fellous TG, McDonald SA, Burkert J, Humphries A, Islam S, et al. (2009) A methodological approach to tracing cell lineage in human epithelial tissues. Stem Cells 27: 1410–1420.

- McDonald SA, Greaves LC, Gutierrez-Gonzalez L, Rodriguez-Justo M, Deheragoda M, et al. (2008) Mechanisms of field cancerization in the human stomach: the expansion and spread of mutated gastric stem cells. Gastroenterology 134: 500–510.
- Shin MG, Kijigaya S, McCoy JP Jr, Levin BC, Young NS (2004) Marked mitochondrial DNA sequence heterogeneity in single CD34+ cell clones from normal adult bone marrow. Blood 103: 553–561.
- Shin MG, Kijigaya S, Tarnowka M, McCoy JP Jr, Levin BC, et al. (2004) Mitochondrial DNA sequence heterogeneity in circulating normal human CD34 cells and granulocytes. Blood 103: 4466–4477.
- Greaves LC, Preston SL, Tadrous PJ, Taylor RW, Barron MJ, et al. (2006) Mitochondrial DNA mutations are established in human colonic stem cells, and mutated clones expand by crypt fission. Proc Natl Acad Sci U S A 103: 714– 719.
- Nooteboom M, Johnson R, Taylor RW, Wright NA, Lightowlers RN, et al. (2010) Age-associated mitochondrial DNA mutations lead to small but significant changes in cell proliferation and apoptosis in human colonic crypts. Aging Cell 9: 96–99.
- Sharpless NE, DcPinho RA (2007) How stem cells age and why this makes us grow old. Nat Rev Mol Cell Biol 8: 703–713.
- Kujoth GC, Hiona A, Pugh TD, Someya S, Panzer K, et al. (2005) Mitochondrial DNA mutations, oxidative stress, and apoptosis in mammalian aging. Science 309: 481–484.
- Trifunovic A, Wredenberg A, Falkenberg M, Spelbrink JN, Rovio AT, et al. (2004) Premature ageing in mice expressing defective mitochondrial DNA polymerase. Nature 429: 417–423.
- Ahlqvist KJ, Hamalainen RH, Yatsuga S, Uutela M, Terzioglu M, et al. (2012) Somatic progenitor cell vulnerability to mitochondrial DNA mutagenesis underlies progeroid phenotypes in Polg mutator mice. Cell Metab 15: 100–109.
- Chen ML, Logan TD, Hochberg ML, Shelat SG, Yu X, et al. (2009) Erythroid dysplasia, megaloblastic anemia, and impaired lymphopoiesis arising from mitochondrial dysfunction. Blood 114: 4045–4053.
- Fox RG, Magness S, Kujoth GC, Prolla TA, Maeda N (2012) Mitochondrial DNA polymerase editing mutation, PolgD257A, disturbs stem-progenitor cell cycling in the small intestine and restricts excess fat absorption. Am J Physiol Gastrointest Liver Physiol 302: G914–924.
- Norddahl GL, Pronk CJ, Wahlestedt M, Sten G, Nygren JM, et al. (2011) Accumulating mitochondrial DNA mutations drive premature hematopoietic aging phenotypes distinct from physiological stem cell aging. Cell Stem Cell 8: 499–510.
- Ericson NG, Kulawiec M, Vermulst M, Sheahan K, O'Sullivan J, et al. (2012) Decreased mitochondrial DNA mutagenesis in human colorectal cancer. PLoS Genet 8: e1002689.
- Vermulst M, Bielas JH, Kujoth GC, Ladiges WC, Rabinovitch PS, et al. (2007) Mitochondrial point mutations do not limit the natural lifespan of mice. Nat Genet 39: 540–543.
- Greaves LC, Beadle NE, Taylor GA, Commane D, Mathers JC, et al. (2009) Quantification of mitochondrial DNA mutation load. Aging Cell 8: 566–572.
- Vijg J (2000) Somatic mutations and aging: a re-evaluation. Mutat Res 447: 117–135.
- He Y, Wu J, Dressman DC, Iacobuzio-Donahue C, Markowitz SD, et al. (2010) Heteroplasmic mitochondrial DNA mutations in normal and tumour cells. Nature 464: 610–614.
- Payne BA, Wilson IJ, Yu-Wai-Man P, Coxhead J, Deehan D, et al. (2013) Universal heteroplasmy of human mitochondrial DNA. Hum Mol Genet 22: 384–390.
- Samuels DC, Li C, Li B, Song Z, Torstenson E, et al. (2013) Recurrent tissuespecific mtDNA mutations are common in humans. PLoS Genet 9: e1003929.

- Brandon MC, Lott MT, Nguyen KC, Spolim S, Navathe SB, et al. (2005) MITOMAP: a human mitochondrial genome database - 2004 update. Nucleic Acids Res 33: D611–613.
- Yao YG, Bandelt HJ, Young NS (2007) External contamination in single cell mtDNA analysis. PLoS One 2: e681.
- Noah TK, Donahue B, Shroyer NF (2011) Intestinal development and differentiation. Exp Cell Res 317: 2702–2710.
- Shoubridge EA (2000) Mitochondrial DNA segregation in the developing embryo. Hum Reprod 15 Suppl 2: 229–234.
- 34. Greaves LC, Elson JL, Nooteboom M, Grady JP, Taylor GA, et al. (2012) Comparison of mitochondrial mutation spectra in ageing human colonic epithelium and disease: absence of evidence for purifying selection in somatic mitochondrial DNA point mutations. PLoS Genet 8: e1003082.
- Greaves LC, Barron MJ, Plusa S, Kirkwood TB, Mathers JC, et al. (2010) Defects in multiple complexes of the respiratory chain are present in ageing human colonic crypts. Exp Gerontol 45: 573–579.
- Bandy B, Davison AJ (1990) Mitochondrial mutations may increase oxidative stress: implications for carcinogenesis and aging? Free Radic Biol Med 8: 523– 539.
- Coller HA, Khrapko K, Bodyak ND, Nekhaeva E, Herrero-Jimenez P, et al. (2001) High frequency of homoplasmic mitochondrial DNA mutations in human tumors can be explained without selection. Nat Genet 28: 147–150.
- Elson JL, Samuels DC, Turnbull DM, Chinnery PF (2001) Random intracellular drift explains the clonal expansion of mitochondrial DNA mutations with age. Am J Hum Genet 68: 802–806.
- Kowald A, Kirkwood TB (2013) Mitochondrial mutations and aging: random drift is insufficient to explain the accumulation of mitochondrial deletion mutants in short-lived animals. Aging Cell 12: 728–731.
- Ameur A, Stewart JB, Freyer C, Hagstrom E, Ingman M, et al. (2011) Ultradeep sequencing of mouse mitochondrial DNA: mutational patterns and their origins. PLoS Genet 7: e1002028.
- Greaves LC, Barron MJ, Campbell-Shiel G, Kirkwood TB, Turnbull DM (2011) Differences in the accumulation of mitochondrial defects with age in mice and humans. Mech Ageing Dev 132: 588–591.
- Kennedy SR, Salk JJ, Schmitt MW, Loeb LA (2013) Ultra-sensitive sequencing reveals an age-related increase in somatic mitochondrial mutations that are inconsistent with oxidative damage. PLoS Genet 9: e1003794.
- Williams SL, Mash DC, Zuchner S, Moraes CT (2013) Somatic mtDNA mutation spectra in the aging human putamen. PLoS Genet 9: e1003990.
- Stewart JB, Freyer C, Elson JL, Wredenberg A, Cansu Z, et al. (2008) Strong purifying selection in transmission of mammalian mitochondrial DNA. PLoS Biol 6: e10.
- Fan W, Waymire KG, Narula N, Li P, Rocher C, et al. (2008) A mouse model of mitochondrial disease reveals germline selection against severe mtDNA mutations. Science 319: 958–962.
- Ross JM, Stewart JB, Hagstrom E, Brene S, Mourier A, et al. (2013) Germline mitochondrial DNA mutations aggravate ageing and can impair brain development. Nature 501: 412–415.
- Finch C, Kirkwood TBL. (2000) Chance, Development and Ageing: Oxford University Press.
- Langie SA, Lara J, Mathers JC (2012) Early determinants of the ageing trajectory. Best Pract Res Clin Endocrinol Metab 26: 613–626.
- Bigger BW, Liao AY, Sergijenko A, Coutelle C (2011) Trial and error: how the unclonable human mitochondrial genome was cloned in yeast. Pharm Res 28: 2863–2870.
- Seneca S, Vancampenhout K, Van Coster R, Smet J, Lissens W, et al. (2014) Analysis of the whole mitochondrial genome: translation of the Ion Torrent Personal Genome Machine system to the diagnostic bench? Eur J Hum Genet. E-pub ahead of print. doi:10.1038/ejhg.2014.49.